Author's response to reviews

Title: Multilevel analysis of HIV related risk behaviors among heroin users in a low prevalence population

Authors:

  Huizhen Li (fallenkerry@hotmail.com)
  William Goggins (wgoggins@cuhk.edu.hk)
  Shui Shan Lee (sslee@cuhk.edu.hk)

Version: 6 Date: 28 April 2009

Author's response to reviews: see over
FORMATTING CHANGES

Major revision

Title and author list – checked

Conclusion The last section has been revised, with the addition of XONCLUSION on page 15.

References – this section has been revised. The specific changes are:
  o The references are in order
  o Reference 2 has been replaced
  o References 4, 5, 6, 12, 25 31 and 34 have been revised as suggested.
  o Tables – The tables are now in separate files. A section on the legends of the tables has been added.
  o Figure legends – this section has been revised accordingly.

Minor revision

Keywords after Abstract have been deleted.

Email addresses of all authors are now on title page.

Underlining in the Abstract have been removed.

INTRODUCTION has been replaced by BACKGROUND.

Final proofreading has been performed.